Author's response to reviews

Title: Heparan sulfate proteoglycans undergo differential expression alterations in right sided colorectal cancer, depending on their metastatic character

Authors:

Iván Fernández-Vega (ivan_fernandez_vega@hotmail.com)
Olivia García-Suárez (garciaolivia@uniovi.es)
Beatriz García (garciabeatriz@uniovi.es)
Ainara Crespo (ainaracsu@gmail.com)
Aurora Astudillo (astudillo@hca.es)
Luis M Quirós (quirosluis@uniovi.es)

Version: 4
Date: 4 June 2015

Author's response to reviews:

Editorial Office
BMC CANCER

Dear Editors:

Please find enclosed a revised version of our manuscript “Heparan sulfate proteoglycans undergo differential expression alterations in right sided colorectal cancer, depending on their metastatic character”, in which we have included the modifications suggested in your letter (highlighted in red in the text).

Yours faithfully,

Dr. Luis M. Quirós
Departamento de Biología Funcional. Facultad de Medicina.
C/Julián Clavería s/n, 33006-Oviedo, SPAIN.
Tel.: 985103000 ext.: 5213
E-mail: quirosluis@uniovi.es